Author’s response to reviews

Title: Synthetic Fiber from a Teddy Bear Causing Keratitis and Conjunctival Granuloma: Case report

Authors:

Mohammed K Farooq MKF (kashfaro@gmail.com)
Jan U Prause JUP (jup@sund.ku.dk)
Steffen Heegard SH (sthe@sund.ku.dk)

Version: 3 Date: 12 June 2011

Author’s response to reviews:

To the Chief Editor of BMC ophthalmology, 11.06.2011

We kindly thank for the acceptance of the manuscript.

The latest recommended changes have been made and a point-by-point description is given below.

1. Authors’ names are corrected as suggested.

2. The structure of the abstract has been reformatted to the following: Background, Case presentation, Conclusions.

3. The structure of the manuscript has been reformatted to the following: Background, Case presentation, Conclusions, Competing interest (with funding), authors’ contributions, Acknowledgement (with consent), references and legend.

4. The highlighting of the text has been removed.

5. All figures are cited in the manuscript.

With kind regards,

Prof. Steffen Heegaard
Eye Pathology Section, University of Copenhagen, Denmark.